Author's response to reviews

Title: Lack of associations between two previously identified susceptible single nucleotide polymorphisms of Interleukin-23 receptor gene and ankylosing spondylitis: a replication study in a Chinese Han population

Authors:

Bang-ping Qian M.D. (qianbangping@163.com)
Jun Jiang Ph.D. (spine821107@gmail.com)
Ming-liang Ji Ph.D. (823305196@qq.com)
Bin Wang M.D. (scoliosis2002@sina.com)
Yang Yu M.D. (scoliosis2002@sina.com)
Yong Qiu M.D. (scoliosis2002@sina.com)

Version: 7 Date: 13 May 2013

Author's response to reviews: see over
Dear Editor:

Thanks a lot for the comprehensive review of our manuscript entitled “Lack of associations between two previously identified susceptible single nucleotide polymorphisms of Interleukin-23 receptor gene and ankylosing spondylitis: a replication study in a Chinese Han population” (MS: 1255460407843368). We highly appreciate that you will in principle accept our paper. Following your suggestion, our paper has been edited by Edanz (www.edanzediting.com/bmc1) to improve the language. Here we submit the revised manuscript and hope that the revised manuscript will finally meet your requirements for publication.

Many thanks to you, the editor and the reviewers, for the time and effort spent on this paper.

Yours sincerely,
Yong Qiu